Author's response to reviews

Title: Can the feedback of patient assessments, brief training, or their combination, improve the interpersonal skills of primary care physicians? A systematic review

Authors:

Sudeh Cheraghi-Sohi (sudeh.cheraghi-sohi@manchester.ac.uk)
Peter Bower (peter.bower@manchester.ac.uk)

Version: 5 Date: 20 August 2008

Author's response to reviews:

No changes were requested by the reviewer. The formatting change required by the journal have now been made.